Author's response to reviews

Title: Factors influencing the development of evidence-based practice among nurses: a self-report survey

Authors:

Anne Dalheim (anne.dalheim@helse-bergen.no)
Stig Harthug (stig.harthug@helse-bergen.no)
Roy M Nilsen (roy.miodini.nilsen@helse-bergen.no)
Monica W Nortvedt (monica.nortvedt@hib.no)

Version: 2 Date: 8 January 2012

Author's response to reviews:

Anne Dalheim,
Nurse and Msc in Evidence-practice
Centre for Evidence Based Practice,
Bergen University College,
Bergen, Norway
January 8, 2012
Dear Editor of BMC Health Services Research,

I will inform you that I have updated the manuscript on the following points:

1. I have clarified that the Privacy ombudsman at Haukeland University Hospital has approved the project.
2. I have included declaration of competing interests.
3. I have added the Authors’ Contribution section after the Competing section.

I am sorry that the manuscript did not include these points at the first submission.

Sincerely,
On behalf of authors,
Anne Dalheim